**Author’s response to reviews**

**Title:** Community perspectives on randomisation and fairness in a cluster randomized controlled trial in Zambia

**Authors:**

Maureen Kombe (kombe68@yahoo.com)

Joseph Zulu (josephmumbazulu@gmail.com)

Charles Michelo (ccmichelo@yahoo.com)

Ingvild Sandøy (ingvild.sandoy@uib.no)

**Version:** 3  **Date:** 23 Oct 2019

**Author’s response to reviews:**

Dear editors,

Thank you very much for pointing out some technical omissions. We have done as advised.

1. **Title page**

Please indicate the corresponding author on your title page and provide their email address.

Response: This has now been done.

2. **Consent for publication**

We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Response: This has now been done.

3. **Funding**

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: This has now been done.